Author’s response to reviews

Title: Prehospital transportation to therapeutic hypothermia centers and survival from out-of-hospital cardiac arrest

Authors:

Derek DeLia (ddelia@ifh.rutgers.edu)
Henry E Wang (hwang@uabmc.edu)
Jared Kutzen (jkutzin@gmail.com)
Mark Merlin (MMerlin@barnabashealth.org)
Jose Nova (jnova@ifh.rutgers.edu)
Kristen Lloyd (klloyd@ifh.rutgers.edu)
Joel C Cantor (jcantor@ifh.rutgers.edu)

Version: 2 Date: 7 March 2014

Author’s response to reviews:

Dear Editors,

Please find the requested revisions regarding informed consent and table formats.

Sincerely,

Derek DeLia